Accelerator Status Report

Run Schedule: Aug 21, 2009 to Dec 23, 2009
Nov. 18, 2009

Injector

· The experts spent more than two shifts to troubleshoot the high Hall A Compton noise problem. The beam was centered in the 5MeV dipole and the vertical BPM offsets were adjusted in 0L and 0R regions, which eliminated the beam scraping and reduced the Compton noise.  

· Injector was optimized to run up to 170uA to the inline dump. Three beams do not coincide, which should be addressed as soon as possible.
Accelerator   Energy: 6.068 GeV
· Overall the accelerator was running well with high current (110uA) to Hall A.
· We did a spin dance to determine a best Wien angle for the halls.

· RSEP8A tripped off a couple of time due to phase unbalance fault. 

· RF 1L04 trips off HV. EES replaced the Klystron #2 body return LCW hose.
· There is a 0.4 - degree offset in the linac gang phase PID locks. This is part of the solution to lower the Compton noise level.

· Thursday swing shift the storm caused a power glitch, which caused a couple of hours’ downtime.
· MAN5R04 mismatches. After a down beam got stuck in 5R due to the shun mismatch. When acknowledging an alarm make sure you understand it otherwise you mask a real problem.
· Some trim card replaced.

· MYA8T01 mismatched and the power supply was replaced.

· Hall A dipole tripped off a couple of times due DC overload.

· XSEP2T tripped off twice due to phase unbalance faults.

· Some iocs like iocha1 does not have Save/Restore button. Those iocs do not restore.
· LEM parameters: the default current is 500uA; we should put 600uA for current run.

· The accelerator was stable after the energy change. The accelerator experts continue to troubleshoot the high Compton noise problem.
· Overall RF was doing well with the high energy run. 
· It appears that something is going on with the MO in the NL linac which caused the NL RF gang phase to shift twice and several cavities to trip each time. EES will investigate the problem.
· 1L06 insulating vacuum pump trip off, which caused the liquid level to go down. The pump was turned back on.

· 5pass CW beam to BSY was set up. We can deliver 5uA CW beam to the BSY dump as a witness beam while Hall A is down.

· Arc4 GOFs were adjusted to get Dogleg4 out of rail. 
· IHA1C05 was fixed.

· Two jumpers were installed: Hall C line viewer limited mode was jumped due to installation of Qweak; Hall C BCM was also jumpered due to the installation work.
· Thursday (11/05) five pass separation drifted, which caused BLMAT07 to trip. The separation was optimized.  Last night (11/10) swing shift the separation drifted again which caused Hall B BOM counts to go high.  
Hall A---PVDIS; pass 5; energy 6.068 GeV
· After the optimization of the injector the Compton noise was much reduced. The tuning of the injector is done. The hall will check the Compton on their side. They already added some shielding to its photon detector, which appeared to help some.
· Hall A Moller procedure file download: dispersive optics vs. Moller optics. A wrong optics was downloaded a couple of times. During Hall A Moller runs one and only one of the three locks should be on: Arc2 energy lock, Hall C FFB on in energy mode, Hall A energy PID lock. The procedure is being revised to make it better. 
· MST software code was modified for Hall A energy measurement to make sure the energy measurement is done in straight ahead mode. The MST now sets the Compton dipole to zero when downloading the dispersive optics.

· Hall A have problems with its Moller target. Sometimes they cannot move the target.

· Hall A target chamber valve closed a few times. The trip level was adjusted. 
· Hall A Compton noise level is too high. The accelerator experts have been                  troubleshooting the problem and have not identified the root cause yet. 
· Hall A has taken up to 110uA beam and beam is stable.

Hall B ---Eg6; pass 5; energy 6.068 GeV
· Hall B current lock did not take the hall attenuator to the set level a couple of times after trips. So beam come back hot and tripped the hall detector HV. 

· When adjusting Hall B beam profile use ITV2C23. Making a round spot (by adjusting 2C04 and 05 quads) on ITV2C23 will make beam round at 2C24. Then use 2C20 and 2C21 quads to adjust beam at 2H00. 
Hall C---HKS; pass 2; energy 2.344 GeV
· Hall C is down for installation of Qweak. It will be back at the end of May in 2010.
· HKS ended successfully on Nov. 2.

